Author’s response to reviews

Title: Problem-solving therapy for depression and common mental disorders in Zimbabwe: piloting a task-shifting primary mental health care intervention in a population with a high prevalence of people living with HIV

Authors:

Dixon D Chibanda (dichi@zol.co.zw)
Petra P Mesu (petra.mesu@zol.co.zw)
Lazarus L Kajawu (lkajawu@gmail.com)
Frances F Cowan (f.cowan@ucl.ac.uk)
Ricardo R Araya (r.araya@bris.ac.uk)
Melanie M A Abas (melanie.abas@kcl.ac.uk)

Version: 8 Date: 19 October 2011

Author’s response to reviews: see over
Dear Editor

We thank you for accepting our paper for publication. We have addressed the formatting issues that you raised, including order of the authors.

Thank you

Dr. D Chibanda